Author's response to reviews

Title: Indicators and measurement tools for health systems integration: A knowledge synthesis protocol

Authors:

Nelly D. Oelke (nelly.oelke@ubc.ca)
Esther Suter (esther.suter@albertahealthservices.ca)
Maria Alice Lima (malice@enf.ufrgs.br)
Cheryl Van Vliet-Brown (cheryl.vvbrown@ubc.ca)

Version: 2 Date: 2 April 2015

Author's response to reviews:

As per the request of the editor, we have added the acceptance letter for the grant application and the ethics approval.